Author’s response to reviews

Title: Combined immune checkpoint inhibitor therapy with nivolumab and ipilimumab causing acute-onset type 1 diabetes mellitus following a single administration: two case reports

Authors:

Marco Zezza (Marco.Zezza@unisante.ch)
Christophe Kosinski (christophe.kosinski@chuv.ch)
Carine Mekoguem (carine.mekoguem@chuv.ch)
Laura Marino (laura.marino@chuv.ch)
Haithem Chtioui (haithem.cthioui@chuv.ch)
Nelly Pitteloud (nelly.pitteloud@chuv.ch)
Faiza Lamine (faiza.lamine@chuv.ch)

Version: 1 Date: 03 Nov 2019

Author’s response to reviews:

Dear Dr Ferreira-Hermosillo,

My co-authors and I are very grateful for the opportunity to submit a revised version of our manuscript. We are also very grateful to the three expert reviewers whose constructive comments have helped us to improve the manuscript.

As requested, we have provided a point-by-point response letter to the reviewers' comments, indicating the respective changes we have made in the manuscript. We also ensured that all changes to the manuscript are indicated in the text using track changes.

We have to inform you and the reviewers, that the line numbers indicated in the table are based on the manuscript in track changes mode with all the markups visible on the right side.

Thank you very much for you kind consideration

With best regards,
Faiza Lamine